**Author's response to reviews**

**Title:** The sensitivity of real-time PCR amplification targeting invasive Salmonella serovars in biological specimens

**Authors:**

Tran VT Nga (ngatvt@oucru.org)  
Abhilasha Karkey (abhilashakarkey@hotmail.com)  
Sabina Dongol (dongolsabina@yahoo.com)  
Hang N Thuy (hangnt@oucru.org)  
Sarah Dunstan (sdunstan@oucru.org)  
Kathryn Holt (kh2@sanger.ac.uk)  
Le TP Tu (tultp@oucru.org)  
James I Campbell (jcampbell@oucru.org)  
Tran T Chau (chauvt@oucru.org)  
Nguyen VV Chau (chaunvv@oucru.org)  
Amit Aryjal (amitarjyal@yahoo.com)  
samir koirala (samir_koirala@yahoo.com)  
Buddha Basnyat (rishibas@wlink.com.np)  
Christian Dolecek (cdolecek@oucru.org)  
Jeremy Farrar (jfarrar@oucru.org)  
Stephen Baker (sbaker@oucru.org)

**Version:** 4  **Date:** 27 April 2010

**Author's response to reviews:**

Oxford University Clinical Research Unit  
Hospital for Tropical Diseases  
190 Ben Ham Tu, Quan 5  
Ho Chi Minh City  
Viet Nam  
sbaker@oucru.org  
Tel: (+84-8) 9 241 761  
Fax: (+84-8) 9 238 904  
17th April 2010

Dear Editor:

Enclosed is the modified manuscript entitled “The sensitivity of real-time PCR amplification targeting invasive Salmonella serovars in biological specimens” by Nga et al. This has been edited as recommended we wish this modified article to be considered for publication in your journal.
This version manuscript has been read and approved by all co-authors in its current form. We think you will find that the concepts of this article are novel, entirely consistent scope of your Journal and will be of significant interest to your readership. We hope you will consider publishing this article.

Thank you for your attention to this matter.

Yours truly,

Dr Stephen Baker